Author's response to reviews

Title: Comparison of the epidemiology and co-morbidities of heart failure in the pediatric and adult populations: a retrospective, cross-sectional study

Authors:
- Gregory Webster (gregory_webster@yahoo.com)
- Jie Zhang (jiezhang00@yahoo.com)
- David N Rosenthal (davidnr@stanford.edu)

Version: 3 Date: 16 May 2006

Author's response to reviews:
We have made all requested changes, as listed below:
1. Letter to editor removed from page 1 of manuscript.
2. Title capitalization fixed.
3. Results and Discussion changed to Results and discussion.
4. Authors' Contributions changed to Authors' contributions, and note added to this section.
5. Page 8, para 2: principle changed to principal.